Author's response to reviews

Title: Reversal of type 1 diabetes via islet beta cell regeneration following immune modulation by cord blood-derived multipotent stem cells

Authors:

Yong Zhao (yongzhao@uic.edu)
Zhaoshun Jiang (zhaoshunjiang@yahoo.cn)
Tingbao Zhao (tingbaozhao@yahoo.cn)
Mingliang Ye (mingliangy@yahoo.cn)
Chengjin Hu (chengjinhu@yahoo.cn)
Zhaohui Yin (yinzhaohui@yahoo.cn)
Heng Li (chinaliheng@yahoo.com.cn)
Ye Zhang (tianhefly2010@yahoo.cn)
Yalin Diao (ydiak@yahoo.cn)
Yunxiang Li (yunxiangli@yahoo.cn)
Yingjian Chen (yinjianchen@yahoo.cn)
Xiaoming Sun (xsun@yahoo.cn)
Mary Beth Fisk (sat@uic.edu)
Randal Skidgel (rskidgel@uic.edu)
Mark Holterman (rmasjet@uic.edu)
Bellur Prabhakar (bprabhak@uic.edu)
Theodore Mazzone (tmazzone@uic.edu)

Version: 2 Date: 28 October 2011

Author's response to reviews:

Dear Dr. Lee:

Please find enclosed the revision of manuscript (MS: 7152996396091117) “Reversal of Type 1 Diabetes via Islet beta Cell Regeneration Following Immune Modulation by Cord Blood-Derived Multipotent Stem Cells”.

We have addressed the reviewers’ comments through a point-by-point response. Please see the Letter-Answers to Reviewers. Based on their constructive critiques and guiding, we have revised our manuscript. All changes have been highlighted in the revision.

Thank you for your kind consideration of our paper.

Sincerely yours,

Yong Zhao, M.D., Ph.D.
Assistant professor
Section of Endocrinology, Diabetes & Metabolism
Department of Medicine
University of Illinois at Chicago
Phone: 312 996 7989
E-mail: yongzhao@uic.edu